Author’s response to reviews

Title: Body weight in midlife and long-term risk of developing heart failure - a 35-year follow-up of the Primary Prevention Study in Gothenburg, Sweden

Authors:

Lena Björck (lena.bjorck@gu.se)
Masuma Novak (masuma.novak@gu.se)
Maria Schaufelberger (maria.Schaufelberger@vgregion.se)
Kok W Giang (wai.giang.kok@gu.se)
Annika Rosengren (annika.rosengren@gu.se)

Version: 3
Date: 23 May 2014

Author’s response to reviews:

Dear Editor,

Please see the revised version of our the paper “Body weight in midlife and the lifetime risk of developing heart failure - a 35-year follow-up of the Primary Prevention Study in Gothenburg, Sweden” where line and page number have been added.

Lena Björck